vc/15-Sep-03
Week 38 starting 15 Sept 03

SM18 Operation SCHEDULE Meeting (Tuesdays)

Schedule for this week : 
	BENCH
	CURRENT

MAGNET
	STATUS 

 Today(15/09)
	Forward look,   & Next Magnet 

	TBA1
	2012
	Under Connection
	Cold by Thursday 18th Sept for cold tests

	TBA2
	3028
	Cold tests completed
	 Disconnect on Wed 17th(?) if warm & Bring 2021

	TBC1
	3046
	To disconnect: had 2 Thrml Cycles
	New magnet to be brought in; work in a stndby to work in parallel with SSS on C2

	TBC2
	SSS1
	Warm 
	Under first time Connection and associated problems

	TBE1
	3043
	Cold 
	Poor Training so far  -needs Extended tests

	TBE2
	1031
	Warm since Friday 12th
	Disconnect &   Bring  New 

2010 

	TBF1
	2025 after F2 done? 
	
	Prepare For 

Commissioning AFTER F2

	TBF2
	2025 proposed
	
	Prepare For 

Commissioning FIRST


 [Note:  NEXT MAGNET IN PIPELINE : 2002 ]

· Test Coordinator this week:  Marco MTA Technician on call:  (16-3602) : 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
· Problems from Operation Last week (Stephane + Test Supervisor )
	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Tuesday 9_M

Tuesday 9_A
	TBE 2 : TEMA not responding during SSL at 4.4 K.

Sun IAS08 : login problem.

Config file power disappeared.
	AB/CO
	TEMA
	Solved.

	Tuesday 9_A
	TBC1: MM Shaft position problem with 1 motor of the TRU
	MTM
	
	Solved by Ewald.



	Tuesday 9_A
	3046, 1031: problem to insert file in MTF

(step 9)


	EDMS
	MTF
	 E-mail sent to EDMS support.

Problems of rights for type L and context for SM18 OP not defined (typing error) 

Solved.

	Tuesday 9_A
	TBE2: I earth  big (120mA)


	MTM
	
	Current depending on the bench.

Statistic is needed

	Tuesday 9_A
	TBC: IDCCT not working


	SL/PO
	
	Solved.

Apparently!

	Tuesday 9_A
	TBC1: MM  big difference of angles between bwd/fwd rotation in few sectors.

Shafts 23/24 
	MTM
	
	Stephane stopped the measurements: Coordinator informed. Investigation on progress.

	Tuesday 9_A
	TBC1: Funny signal on the coils.  
	MTM
	
	Connector Ap1 B/Ap2 B in the mobile rack inverted.

	Tuesday 9_A

Friday 12 
	TBA2 : Anticryostat values not correct.

TBE1 the same.  
	MTM
	
	Coefficients not updated.

Who has to do this?

	Wednesday 10_A
	TBA2: HV test failed
	MTM
	
	Current leads warmed with a fun. Then OK 

	Wednesday 10_A
	TBA2:  Cryo lost 
	MTM/ACR
	
	Bad contact in RVLB8

	Wednesday 10_A
	Plateau of current at 12850 A (ultimate)

Extension of the file is .u and not .a 
	MTM
	
	

	Wednesday 10_A
	TBA2: Minimum energy at 11850 A cannot be triggered.

Quench at 1 A! during the manipulation in Vbus ext.
	MTM
	
	Equip.support called.

Bad configuration in the card of the safety matrix used for the minimum energy) (all LEDs of this card must be off). 



	Friday 12_M
	MM TBA2 (load line , LHC cycle): Field polarity error
	MTM
	
	Equip.support called.

One sector was in E-coil

And pick up coil in a-coil

Error listed in the table.

Has to be seen during the positioning of the shaft 1.5 kA.

	Friday 12_N
	TBC1 : IDDCT not working (current reading)
	SL/PO
	
	Equp.support called.

A led was switched off. Panel open and a cable was missing.

Solved

	Friday 12_A
	TBE1:PLC error Mag300 K and Mag300 K air
	MTM/ACR
	
	Cryo staff started to purge the magnet as IAP not done.

	Saturday 13_A
	TBA2: 2 Leds of the quench valve card not glowing. 

Quench at 1 A! during the manipulation in Vbus ext.
	MTM
	
	Equip.support called.

Card changed.

Solved

	Sunday14_M
	TBE1: IAP @ cold

T coils OK (shaft not there)

Offsets in V-gas_int and V_gas_ext leads 
	MTM
	
	Offsets minimized but errors still there. 

	Sunday 14_M
	TBE1: Quench at 1.5 kA (all LF and HF) 
	MTM
	
	Trigger one or two cards?

Clarification needed.



	Sunday 14_M
	TBE1: Quench at 1.5 kA in Vbus (repeated two times) 
	MTM
	
	Test coordinator informed.

Origin under investigation.

Test coordinator gives the OK to continue.



	Monday 15_A
	TBE2: IAP @warm Quench heater charged because training quench in TBE1
	MTM
	
	Test coordinator informed.

Test can be done later and switch to active/passive. 




